Author’s response to reviews

Title: Laryngeal Polyp Associated with Reflux Disease- Report of a Case Study

Authors:

Wael Hassan (w.hassan@sr.edu.sa)

Version: 3 Date: 18 Nov 2019

Author’s response to reviews:

- Response to reviewers:
  - Authors can expand on the epidemiology of vocal cord polyps caused by GERD and other causes
    ✷ An additional paragraph was added in the Discussion section.

  - If the histopathology of polyps caused by GERD different than caused by other chronic irritants such as smoke, authors could mention about the histology of other irritants in order to further their case stronger
    ✷ An additional paragraph was added in the Discussion section.